Author's response to reviews

Title: Efficacy of multiparametric telemonitoring on respiratory outcomes in elderly people with COPD: a randomized controlled trial.

Authors:

Claudio Pedone (c.pedone@unicampus.it)
Domenica Chiurco (d.chiurco@unicampus.it)
Simone Scarlata (s.scarlata@unicampus.it)
Raffaele Antonelli Incalzi (r.antonelli@unicampus.it)

Version: 6 Date: 24 January 2013

Author's response to reviews: see over
Dear Editor –

Thank you for your rapid response. We reworded the sentences you indicated, to make it clear that our results apply to our sample and may not be applied to the general COPD population. We also changed the first sentence of the Conclusion to avoid speculations on a possible causal mechanisms for the differences observed between groups.

Best regards

Claudio Pedone